Author’s response to reviews

Title: Factors for Short-term Outcomes in Patients with a Minor Stroke: Results from China National Stroke Registry

Authors:

Lingyun Wu (15133916062@163.com)

Anxin Wang (anxin0907@163.com)

Xianwei Wang (xianweiw2005@gmail.com)

Xingquan Zhao (zxq@vip.163.com)

Chunxue Wang (snowsen@126.com)

Liping Liu (lipingsister@gmail.com)

Huaguang Zheng (zhgdoc@163.com)

Yongjun Wang (yongjunwang1962@gmail.com)

Yibin Cao (yibin07@sina.com)

Yilong Wang (yilong528@gmail.com; yilong528@aliyun.com)

Version: 2 Date: 23 Nov 2015

Author’s response to reviews:

Response to Editor

EDITORIAL REQUEST:

- Please add the title name of conclusion section according to BMC Guidelines: http://www.biomedcentral.com/bmcneurol/authors/instructions/researcharticle

Answer: Thank you very much for your comment. We are sorry for our carelessness. As recommended, we have added the title name of conclusion section according to BMC Guidelines. (Highlighted in Page 14)